In response to your E-mail 30 Mar 2006 concerning the manuscript "The Effect of Posterior Subtenon Methylprednisolone Acetate in the Refractory Macular Edema: prospective nonrandomized interventional case series".

1. Additional files were eliminated; because tables included in main manuscript.

2. Acknowledgment was changed to Acknowledgments.

3. Authors' contributions were corrected.

Yours sincerely

AR Javadzadeh

Assistant professor of ophthalmology, Tabriz University of Medical Science, Tabriz, Iran